Title: A Pilot Survey of Post-Deployment Health Care Needs in Small Community-Based Primary Care Clinics

Authors:

Polly H Noel (noelp@uthscsa.edu)
John E Zeber (John.Zeber@va.gov)
Mary J Pugh (pughm@uthscsa.edu)
Erin P Finley (finleye@uthscsa.edu)
Michael L Parchman (parchman@uthscsa.edu)

Version: 4 Date: 25 July 2011

Author's response to reviews:

July 25, 2011

Editors

BioMed Central – BMC Family Practice

Re: submission of reformatted manuscript #6908626365026469

Dear BMC Family Practice Editors:

I apologize for the continued discrepancies in the title and abstract for our manuscript: A Pilot Survey of Post-Deployment Health Care Needs in Small Community-Based Primary Care Clinics for publication in BMC Family Practice. We have made the requested revisions, as outlined below:

Title: please ensure that the title is identical in the manuscript and the submission system

The titles in the manuscript and submission system are now identical.

Abstract: This should be identical in the manuscript and the submission system

The abstracts in the manuscript and submission system are now identical.

Please let us know if you require any further changes. Thank you for your patience.
Sincerely yours,

Polly Hitchcock Noël, PhD (Corresponding Author)

Associate Director, VERDICT, South Texas Veterans Health Care System

Associate Professor, Dept. of Medicine, UT Health Science Center at San Antonio

7400 Merton Minter Blvd, San Antonio, Texas 78229

210-617-5314 (phone)

210-567-4423 (fax)

NOELP@UTHSCSA.EDU